Author's response to reviews

Title: A Survey of Confidence Level and Education of Peripheral Nerve Blocks in Graduating Anesthesia Residents

Authors:

Tiffany Sun Moon (Tiffany.Moon@UTsouthwestern.edu)
Eunjung Lim (Lime@purdue.edu)
Sakura Kinjo (kinjos@anesthesia.ucsf.edu)

Version: 3 Date: 28 September 2012

Author's response to reviews:

Competing interest added.
Authors' contributions added.
Questionnaire attached.
One of the author's contact information changed (married and relocated, Tiffany Sun Moon).